Author’s response to reviews

Title: The 2018 Lebanese Society of Infectious Diseases and Clinical Microbiology Guidelines for the Use of Antimicrobial Therapy in Complicated Intra-Abdominal Infections in the Era of Antimicrobial Resistance

Authors:

Nicholas Haddad (hadda1ne@cmich.edu)
Souha Kanj (sk11@aub.edu.lb)
Lyn Awad (lyn.awad@lau.edu)
Dania Abdallah (rpdania@gmail.com)
Rima Moghnieh (moghniehrima@gmail.com)

Version: 2 Date: 14 Feb 2019

Author’s response to reviews:

All editorial comments were taken in to consideration and the necessary changes were done in the manuscript.